

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Schnellenbach, Jan

Working Paper

The concept of Ordnungspolitik: Rule-based economic policy-making from the perspective of the Freiburg School

Freiburger Diskussionspapiere zur Ordnungsökonomik, No. 21/7

Provided in Cooperation with:

Institute for Economic Research, University of Freiburg

Suggested Citation: Schnellenbach, Jan (2021): The concept of Ordnungspolitik: Rule-based economic policy-making from the perspective of the Freiburg School, Freiburger Diskussionspapiere zur Ordnungsökonomik, No. 21/7, Albert-Ludwigs-Universität Freiburg, Institut für Allgemeine Wirtschaftsforschung, Abteilung für Wirtschaftspolitik und Ordnungsökonomik, Freiburg i. Br.

This Version is available at: https://hdl.handle.net/10419/234118

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.







The Concept of Ordnungspolitik: Rule-Based Economic Policy-Making from the Perspective of the Freiburg School

Jan Schnellenbach

21/7

Freiburger **Diskussionspapiere** zur Ordnungsökonomik

Freiburg **Discussionpapers** on Constitutional Economics

Institut für allgemeine Wirtschaftsforschung Abteilung Wirtschaftspolitik und Ordnungsökonomik

Albert-Ludwigs-Universität Freiburg



The Concept of Ordnungspolitik: Rule-Based Economic Policy-Making from the Perspective of the Freiburg School

Jan Schnellenbach

Brandenburg University of Technology Faculty for Economics, Law and Social Sciences Chair for Microeconomics jan.schnellenbach@b-tu.de

forthcoming in: Public Choice

Abstract. Should economic policy be guided by rules? In this paper, we take the perspective of the Freiburg School and trace its argument for rule-based *Ordnungspolitik* back to the roots of the concept. In doing so, will not offer a comprehensive review of the literature, but argue closely along the works of Walter Eucken, whose works are central to understanding the founding generation of the Freiburg School. We argue that there are costs of not having rules and therefore that the main thrust of the Freiburg approach is still valid. There are good, empirical arguments for pursuing a rule-based Ordnungspolitik in order to avoid the costs of discretionary policy-making. Furthermore, we argue that a reliance on stable rules does not imply an incapacitation of democratic decision-making. Rules rely on democratic support, and rule-based *Ordnungspolitik* also leaves substantial material scope for discretionary democratic decision-making.

Keywords: Ordnungspolitik; Freiburg School; economic orders; economic constitutions.

JEL Codes: B15; B25; B41; H11; P48, P50.

1. Introduction

Should economic policy be guided by rules? Both the constitutional approach of the Virginia School, as well as the ordoliberal approach of the Freiburg School, generally answer that question in the affirmative. But important differences in the details of the respective arguments lead to the conclusion and some differences emerge in the purposes and designs of rules that guide policy. The aim of the paper at hand, within the context of this special issue of *Public Choice*, is to outline the particular position of the Freiburg School on rule-based policymaking, and to trace its argument for rule-based *Ordnungspolitik* back to its historical roots.

In doing so, will not offer a comprehensive review of the entire first generation of the Freiburg School, since that would take us far beyond the scope of a single paper. While different members of the Freiburg School emphasize different issues and arguments, the main thrust of the School's theoretical research program can be extracted from the works of its most prominent proponent, Walter Eucken. Therefore, and because they are essential to understanding the basic arguments of the Freiburg School's founding generation, we will argue closely along the lines laid out in Eucken's work. For a broader overview over the origins of the Freiburg School, see Vanberg (2004) and Kolev (2020). A collection of original texts is assembled in Goldschmidt and Wohlgemuth (2008).

While the German tradition of *Ordnungspolitik*, which originated in the Freiburg School, bears similarities to the constitutional political economy (CPE) of the Virginia School, important differences also can be found. For example, we will show that the concept of economic order is broad and implies more than just formal constitutional rules. Therefore, *Ordnungspolitik* as a policy concept that aims at sustaining a functioning market order includes an emphasis on formal rules, but it also includes discretionary policymaking by design. The Freiburg School ventures away from the normative conviction that a functioning market order is desirable and asks which rule-based and discretionary policies are needed to sustain it. It does, however, not ask explicitly what rule improvements could garner unanimous consent, as the Virginia School does.

Important contributions to the relevant literature argue that key elements of *Ordnungspolitik* in the Freiburg tradition can be reconciled with the modern approach of CPEE and reconstructed as results of constitutional economic reasoning (Vanberg 1988, 2004, 2015; Feld and Köhler 2011). But it is important to reckon that historically, *Ordnungspolitik* started

from a different premise, namely that market competition needs to be sustained by the state and that that function is desirable because a breakdown of market competition has been associated historically with negative outcomes.

In Section 2, we will discuss the concept of economic orders and their relationship to systems of (formal) rules. To avoid a common misunderstanding, it is clarified that economic orders are not determined completely by formal rules but have a broader meaning. In Section 3, we discuss positive theoretical perspectives on the emergence of and changes in economic orders. Section 4 outlines the normative perspective: Which economic order is considered superior by the Freiburg School, and why? Section 5 adopts a more practical perspective and discusses the implementation of a competitive order. Section 6 discusses a critical counterargument stating that rule-based policymaking imposes unwanted restrictions on a democratically elected sovereign. And Section 7 discusses whether some of the core assumptions the Freiburg School made in order to justify rule-based *Ordnungspolitik* remain are applicable to current policy problems. Finally, Section 8 concludes.

2. Economic orders are not (only) systems of rules

When Walter Eucken (1939, p. 37)) discusses economic orders, his analytical starting point has been a variety of organizational forms for economic activity, which he had observed both in time and across places. He illustrates the variety using a number of historical examples for different types of economic organization; in discussing them, he also rejects different established explanations for their emergence. For example, Eucken discusses theories of historical determinism, wherein one type of economic organization necessarily follows another, and presents examples of historical contingencies leading to empirically observable sequences of types of economic organization that were incompatible with historical-determinist theories (ibid., p. 53).

At the same time, Eucken also argues against the concept of styles of economic organization, which would be determined by the specific culture of a region or a country. Again, his criticism is motivated empirically: He discusses historical examples of different places within culturally similar regions being characterized by rather different types of economic organization (Eucken 1939, pp, 60-61). For instance, Eucken compares German cities in the 15th century, wherein differences in relative political power between groups led to differences

in economic organization (ibid.), despite very similar cultural backgrounds. With his rejection of the theories of styles of economic organization and of historical determinism, Eucken pits himself against important theories of the German Historical School of Economics.

Instead of searching for historical or cultural determinants of economic organization, Eucken (1939, p. 62) proposes to look directly at the rules that govern economic activity: "Welche Spielregeln herrschen?" is his central question, or "Which are the governing rules of the game?" The rules of the game, on the other hand, are the object of economic policy. They can be either set or at least influenced through political decisions, and are not the result of unalterable historical or sociological laws. The rules of the game are, however, not just formal rules. The rules need to be interpreted and the dominant interpretation of formal rules also is part of the rules of the game.

Furthermore, different market structures may emerge under similar formal rules, but the market structure, exerting a strong influence on the economic behavior of individual producers and consumers, is itself a part of the economic order (Eucken 1939, p. 218, 1952, p. 22). For example, the same set of formal rules may allow for a competitive oligopoly or a monopoly to emerge. Which of them actually surfaces depends on decisions made by market participants that are not fully determined by the rules, or by pure luck: the outcomes of R&D projects, for instance. In any case, the actual market structure is to be accounted for as part of the actual economic order. Everything that influences how market participants make decisions is, according to Eucken, part of the economic order – even if it may not explicitly be part of a formal economic constitution.

An economic order, according to Eucken, therefore is more than a formal set of rules, because the actual rules of the game are not entirely and perfectly determined by any set of formal rules. He adopts a telling metaphor when he compares economic orders to elaborate pieces of classical music by Bach, Mozart or Beethoven (Eucken 1939, p. 202). He points out that all of those musical compositions rely on a fixed system of musical notes and that the composer's creative effort requires, in principle, nothing more than an arrangement or rearrangement of those given notes. Similarly, he argues that empirical economic orders combine different elements from ideal types of economic orders. The ideal types, Eucken (1952, p. 21) later calls them "reine Formen" (pure forms), simply are central planning on the one side, and coordination of individual plans through a competitive price mechanism on the other. Empirically observed orders, Eucken argued, come in a large variety of different actual arrangements, but are basically different combinations of the two pure forms. For example,

Eucken (1939) argues that empirically, one will find no uniformly planned economy, and also none that uniformly relies on the price mechanism, but always some kind of mixture of the two.

A concise definition of what an economic order actually encompasses is given by Eucken (1952, p. 23): "Die Wirtschaftsordnung eines Landes besteht in der Gesamtheit der jeweils realisierten Formen, in denen Betriebe und Haushalte miteinander verbunden sind, in denen also der Wirtschaftsprozeß in concreto abläuft.", which can be translated as "The economic order of a country consists of all actually realized forms in which the concrete economic process is embedded and that connect firms and households" (own translation). That definition still is somewhat vague, but it clarifies once more that the economic order is about how individual plans are made and coordinated and, moreover, that it would be wrong to equate the economic order with the formal rules that govern an economy. The latter have a more or less immediate impact on the economic order, but they are not the economic order per se.

It is noteworthy that neither that complication, nor the variety of resulting economic orders in reality, led Eucken to advocate a purely descriptive approach to economics, which would serve primarily the purpose of taxonomizing given economic orders. On the contrary, and in strong opposition to the Historical School, his focus is on economic theory to establish a general understanding of the economic and social outcomes associated with observable economic orders. Applied economics, in that sense, involves two steps: Empirically, the task is to discern what kind of economic order we observe in a concrete case; theory is then to be applied to predict the outcomes associated with it (Eucken 1939, pp. 208-216). Furthermore, theoretical reasoning allows one to predict outcomes of policies that attempt to change an economic order. Economic theory is the prerequisite for economic policy on the level of economic orders.

3. The origin of economic orders from a positive perspective

How can the emergence of an economic order be explained? Eucken (1939, pp. 208-216) believed that economic orders normally are not the result of conscious, large-scale design, but rather of processes of evolution. That belief does not imply that deliberate actions do not influence economic orders. On the contrary, Eucken acknowledges that deliberate actions of

individuals within an economic order influence the direction of its further evolution. But such deliberate action already is embedded in a given economic order, i.e., an order never is designed from scratch and path-dependence may be important.

When some interest group uses its political clout to secure privileges, the economic order is influenced, e.g., the misuse of the freedom to contract in order to establish a cartel. An economic order is often the sum of many small, deliberate but uncoordinated changes in the system of rules and, in that sense, the economic order as a whole evolves spontaneously. Furthermore, an economic order evolves through the interaction of formal rules and decisions made within a set of formal rules (Eucken 1939, p. 68). In that respect, Eucken already saw that an economic constitution of formal rules easily may have consequences that were not intended by the rules' authors because individuals often interact with rules creatively, in unanticipated ways.

In his *Die Grundlagen der Nationalökonomie* (Foundations of Economics), Eucken cites many historical examples to illustrate how a close scrutiny of the details of economic organization and market structure at a given place and time can help in understanding the underlying economic order. In discussing these cases, he frequently also presents fragmentary theories of change in economic orders. For example, Eucken (1939, pp. 213-214) outlines how, in Flemish cities of the 13th century, traders of cloth organized and dominated the entire production process of their products, leaving weavers and other craftsmen impoverished. Poverty, in turn, motivated violent protests, and Eucken describes how a change in political power now allowed craftsmen to organize themselves in guilds and, furthermore, how the changed political-economic equilibrium was associated with a change in the economic order, wherein a monopsony of traders and a craftsmen's guild was transformed into a bilateral monopoly.

Eucken and other ordoliberal thinkers certainly have been aware of the fact that an economic order is subject to endogenous change, and that changes in economic orders themselves are a subject worthy of economic analysis. Eucken's sketches of different historical cases involving changes in the economic order are attempts at understanding processes of institutional change by looking for factors that alter the bargaining powers of various social groups. In that respect, the sketches anticipate later, more sophisticated approaches to the issue.

For example, the research program of North (1991, 1994) focuses on institutions as tools that reduce transaction costs, which follow from many different problems such as agency,

measurement, or enforcing property rights. But contrary to, e.g., Williamson (1985), who generally models firm-level institutions as efficient responses to those problems, North puts greater emphasis on political-level institutions and acknowledges that a path-dependent evolution of such institutions does not necessarily lead to efficient outcomes.

North (1994, p. 361) argues that weak competition in political markets can lead to the persistence of inefficient institutions, as can the existence of false perceptions of the relative efficiency of alternative institutions (ibid., p. 363). On the other hand, North uses usury laws as an example for an inefficient institution that eventually broke down in early modern Europe, simply because individuals increasingly became creative in circumventing those laws and the transaction costs of effectively prohibiting positive interest rates became prohibitively high themselves (North 1991, p. 105). And in some cases changes in political bargaining power have been observed to be followed by institutional changes that reduced the discretionary powers of governments, thus securing property rights and paving the way for an expansion of transactions and growth (e.g., North and Weingast 1989).

Eucken himself had already been keenly aware of such interactions between political, social and economic institutions. In analyzing them, he coined the term *interdependency of orders*. That led Eucken (1952, pp.180-184) to discuss different channels of causation of institutional change between political, sociological and economic factors. For example, he argues that a federal political order generally is incompatible with central planning, and that a decision to centralize economic decision-making will lead to an erosion of the federal order. Eucken expects that unchecked monopoly power in a market economy will undermine the rule of law, which in turn has further adverse effects on the efficiency of economic institutions.

A political decision to impose tariffs will, Eucken argues, have a broader impact on the economic order because it increases the market power of firms protected from foreign competition. That argument is rather close to later theories of institutional change that rely on changes in bargaining power between groups as an important trigger, leading Eucken (1952, p. 221) to warn that if they are not analyzed sufficiently, small changes in either the political, the economic or the societal order can be like snowballs that trigger an avalanche, generating and lead to large, unintended consequences in the institutional framework as a whole.

Obviously, Eucken and other early proponents of the Freiburg School could not anticipate later developments in institutional economics and political economy. From the perspective of institutional economics, an efficient change in the economic order or inefficient persistence of

a given economic order, both can be plausible outcomes, depending on relative bargaining powers of different interests, as well as on technological progress and the degree of rationality underlying political choices.¹ In some cases, efficient institutional change likewise can be imposed externally by military intervention (Acemoglu et al. 2011).

For obvious reasons, Eucken lacked the formal theoretical apparatus and the quantitative empirical models that modern public choice and political economics bring to bear. But from analyzing historical cases, he reached the unambiguous conclusion that no reason can be found for a Panglossian attitude towards actual economic orders and that any actual economic order is the result of historical and political contingencies, thereby becoming a possible candidate for improvement through purposeful reform. In clear opposition to a Panglossian view, both Eucken and other first-generation members of the Freiburg School, such as Franz Böhm et al. (1936, p. 31), have argued against fatalism and historical determinism and in favor of an economic science whose ultimate rationale should be to guide the purposeful shaping of the economic order.

In sum, the most important similarity between the Freiburg School and modern approaches of public choice and political economics is that the early Ordoliberals already saw the importance of interdependent causal relations between the economic order, social and political institutions. And what is most important, they did not simply assume that political institutions determine the economic order, but that the latter can itself influence the evolution of the former. That insight has been essential for the Freiberg School's normative perspective on the economic order: If the economic order allows for an unfettered accumulation of power, desirable characteristics of the political order come under threat.

4. Which economic order should rules impose? The normative view

The claim that pure laissez-faire had failed is the analytical starting point of German Ordoliberalism, resulting from the experience of the Great Depression, but also from the undermining of market competition by openly formed cartels before the Great Depression. Laissez-faire in that context is defined as an order wherein governments provide a legal order, in particular guaranteeing private property rights, but do not actively supervise and influence

_

¹ Regarding the latter, also see Denzau and North (1994). For an extreme case of the persistence of inefficient institutions, see Acemoglu et al. (2001).

the evolution of the economic order. In other words, laissez-faire is a regime in which the economic order itself evolves spontaneously within a legal order that protects property rights (Eucken 1952, pp. 26-54). Again, the importance of distinguishing between formal rules and the actual economic order becomes obvious. Eucken reckons that within a given, formal economic constitution, the unguarded evolution of the economic order can run in an unintended, from a normative point of view, inferior direction.

Consequently, the early ordoliberals criticized the unchecked accumulation of economic power, be it under a central planning agency or with monopoly in a market economy (Eucken 1939, p. 239). Regarding the negative effects of market power, Eucken (1947, pp. 145-146) takes a broad perspective. He names a failure of the price mechanism to allocate resources efficiently when prices rise above competitive levels. He also makes an argument akin to rent-seeking, and expects that economic power favors and often enforces protectionist policies. And he makes the political-economic point that the abuse of market power can trigger a chain of discretionary interventionist policies, such as price controls, that eventually threaten the working of the price mechanism fundamentally. But finally, Eucken (ibid.) also adds a rather sociological observation, namely that economic power leads to what he calls "Vermassung", a loss of individual freedom and also of individuality with respect to character, lifestyle, or consumption patterns, owing to a dependency on few, powerful businesses and other organizations.

Contrast that with Eucken (1947, p. 147) characterizing a competitive market economy. Here, he argues, the price mechanism facilitates efficient use of resources. The process of competition enhances performance, rather than setting incentives for destructive behavior, as would prevail in a tournament for monopoly power. An order wherein firms and individuals are accustomed to competition, Eucken argues, also will not easily fall for protectionist temptations. And, finally, a competitive order also preserves individual liberty and autonomy.

That simple comparison of two economic orders by Eucken reveals the criteria that underlie his normative thinking. Efficiency in the allocation of resources is important, as is a process of competition that ultimately improves consumer welfare. But in sharp contrast to welfare economic approaches, individual welfare and autonomy both carry normative weight in their own right. They are important not only instrumentally, e.g., because an individual knows best for herself what she requires to maximize her own welfare, but because they are valuable in themselves. Eucken (1947, p. 150) sums that conclusion up himself, when he characterizes

economics as a science that should help "eine menschenwürdige und funktionsfähige Ordnung zu schaffen", i.e., "to create a humane and functional order" (own translation).

In a similar vein, Franz Böhm (1971, pp. 308-311) warns of the accumulation of political power that occurs when the economic order of a competitive market economy increasingly is crowded out by political interventions and elements of central planning. He argues that the loss of individual autonomy associated with extensive political power over economic planning must seem terrifying, and act as a motivation to seek improvements of the competitive market order instead. What is more important, Böhm (1936, p. 120) explicitly points out that a competitive order will not come about spontaneously, but always depends on the legal framework. Without governments enforcing appropriate rules of the game, a competitive order will neither come into existence, nor be stable once it does.

Economic power is transformed into political power through different channels. The first has been sketched in Section 3, where we discussed Eucken's argument that economic power can lead to undue influence in political bargaining processes over institutional changes. That is essentially an early, simple variant of later theories of interest group influence. The second channel is a bit more sophisticated and has been invented by Böhm. The less a firm is exposed to competitive pressures, the greater scope it has to impose self-conceived, private rules that benefit its owners or their transaction partners. By drawing up contract clauses, it creates its own private law. Only a competitive market order ensures that all potential transaction partners can reject a company's self-conceived private law and look for more beneficial conditions elsewhere.

In some cases, attempts to identify and foster a humane economic order also led to somewhat odd policy proposals from individual ordoliberals. For example, Wilhelm Röpke (1944) argued in favor of deliberate promotion of production technologies that supply competitive advantages to small businesses. It is interesting that Röpke's peculiar proposal involves a deliberately discriminatory intervention, in favor of small firms, even if larger firms do not enjoy significant market power. Such a discriminatory approach runs counter to the general, rule-based approach Ordoliberalism usually proposes, but Röpke deems it necessary in that specific case to restrict production to a small and therefore, in his opinion humane, scale.

The claim that an economic order ought to serve different objectives clearly can lead to contradictions if they need to be traded off against one another. In the example above, Röpke puts great emphasis on what he believes to be a necessary characteristic of humane societies,

namely, the organization of economic activity in small firms rather than large organizations in which the single individual is anonymous and interchangeable. A different ordoliberal economist, with less of a conservative Christian background than Röpke, could emphasize efficiency to a much greater extent, arguing in favor of exploiting economies of scale in larger firms. Given our discussion thus far, it therefore is not entirely clear which goals an ordoliberal, rule-based policy should serve.

Eucken and other members of the Freiburg School, such as Franz Böhm, distinguish themselves from other ordoliberals, like Röpke, by assigning unambiguous priority to securing competition as the material goal of economic policy and implementing a non-discriminatory, privilege-free order to do so (Vanberg 2004, p. 2). However, that recommendation does not simply reflect a personal preference, e.g., for economic efficiency and against having a more humane order (or what some people may believe to be a humane order), but an empirically testable claim that a competitive order that puts restraints on individual power simultaneously helps to achieve both an efficient *and* a humane economic order.

That claim can be made only on the level of general rules. A welfare economist micromanaging an economy always will see tradeoffs between the two objectives, as Röpke likewise did. Owing to his (at least implicit) claim that some sizes of firms optimally balances efficiency concerns with the concern of having a humane order, firm size itself becomes a policy objective. A policymaker would have to leave the level of general, rule-based policies and engage in micro-managing the economy, as a neoclassical welfare maximizer would do.

But on the level of rules, one can even claim that Röpke's optimal intervention rests on a misperception because one cannot pursue isolated, *prima facie* optimal interventions into the market process. The reason reveals a categorical difference between *Ordnungspolitik* and *Prozeßpolitik*, between rule-level policymaking and interventions into the market process, as perceived by Eucken. The former sets (or at least influences) the rules of the game, the latter attempts to steer the economy towards specific desired results by direct intervention. But, as Eucken argues, attempts at *Prozeßpolitik* often lead to spirals of intervention. If policymakers distort one margin, private-sector adaption often will lead to unintended consequences, which beg another intervention, and so on. *Prozeßpolitik*, argues Eucken, itself threatens the stability of a market order and leads to a crowding out of competitive forces by central planning (Vanberg 2004, p. 2). However noble the underlying intentions may be, attempts to directly intervene into the market process, according to Eucken, risk severe damage to the competitive

economic order as a whole and through the interdependency of orders also to the political and societal order. Therefore, economic policymaking always should aim at a rational design of the rules that shape (albeit sometimes indirectly) the economic order and to make sure that that order is a competitive one – because only then is the stability of an order that also is humane secured.

5. Rule-based Ordnungspolitik: a more practical perspective

Eucken was well aware of some issues of the political economy of providing economic policy expertise. He argued that such expertise could well be used for problematic purposes and that even well-intended policy efforts, for example, those directed at reducing unemployment, could be abused to further the goals of special interest groups (Eucken 1947, p. 138). Eucken clearly is not taking the same position adopted later by welfare economics, advocating for economics to give counsel to hypothetical, ideal governments. The Freiburg ordoliberals want to improve actual economic policy and they are aware that in order to do so, they will need to account for the problems of actual policymaking processes. They also were aware of the problem that a rational, rule-based set of policies may not be easy to implement. Eucken (1952, p. 219) even argued that times of crisis might be the most important windows of opportunity wherein fundamental reforms of the economic order could find the necessary political support.

The first core element of the Freiberg School's political program is competition policy. Perfect competition is the ideal, a market structure wherein nobody, neither on the supply nor on the demand side, has significant market power and wherein, therefore, the price mechanism leads to an efficient allocation of resources. Perfect competition is associated with *Leistungswettbewerb*, a form of competition in which suppliers' resources are directed towards improving their goods or lowering their prices, in contrast to more destructive forms of competition, such as actively sabotaging the efforts of rivals in order to gain market power. In a world where rule-based competition policy prohibits firms from harvesting excessive monopoly rents, such destructive efforts appear irrational (Eucken 1952, p. 247). Such a competition policy needs to be predictable; it needs to define clear criteria for a regulatory body to intervene and, for example, prohibit mergers of some companies. That rule-based approach also is a self-commitment of governments to refrain from unwarranted interventions into the market process itself: The less discretionary leeway they have, the less likely are

abuses of the power in the conduct of competition policy, e.g., in order to support special interest groups.

Therefore, when we discuss *Ordnungspolitik* as rule-based policy, the discussion relates to at least two levels. One level is the policy directed at the economic order, which market participants experience as the rules of the game. Such policies can come as negative, formal rules addressed at market participants and also as political efforts to create the necessary conditions for a competitive order (Eucken 1952, p. 255). The second level, however, political self-restraint, implies that constitutional rules restricting the scope of government interventions also are an element of *Ordnungspolitik*. Governments need to be prohibited from implementing policies that endanger the proper functioning of the price mechanism, with everything that entails: no inflationary monetary policies, no protectionism, no favoring of market incumbents, no subsidies that prevent the exit of insolvent firms, no direct price controls, and many more (Ibid.). On the second level, *Ordnungspolitik* likewise always entails the task of looking for rules that ensure governmental self-restraint regarding such undesired policies (also see Böhm 1966).

Eucken (1952, pp. 254-303) is well-known for having established a number of principles that he believed should guide *Ordnungspolitik* and that sum up in a more systematic way what governments should and should not do. His *konstituierende Prinzipien* (constitutive principles) need, in his own view, to be followed in order to establish the conditions for a competitive economic order to be functional at all. They are:

- 1. *Stable money*: Eucken argues that stable money is a prerequisite for the price mechanism to work, and he argues that that goal may require a monetary order that leaves little leeway for the monetary authorities to engage in discretionary policymaking. A monetary constitution that grants technocrats independence and commits them to a stable inflation rate target would, in his view, not be sufficient. Instead, a rule that ensures stable and predictable rates of money supply growth could be desirable (Eucken 1952, p. 259).²
- 2. *Open markets*: Eucken was convinced that a lesson to be learned from economic history is that closure of markets facilitates the emergence of cartels and generally secures market power. That obviously is true for some cases of market closure, such

² For a more extensive discussion of Ordoliberalism and monetary policy, also see Feld et al. (2021) and Köhler and Vanberg (2015).

- as tariffs or regulatory burdens that deter entrants. It often also is true of a denial of market exit, if, for example, inefficient incumbents receive subsidies and thereby deter Schumpeterian processes of creative destruction.³ His focus on openness even led Eucken to be critical of patent laws, which he suspected to guard inefficient market power (Eucken 1952, pp. 268-269).
- 3. Secure private property rights: Anticipating some later developments in the new institutional economics, Eucken saw the importance of having secure private property rights. He substantiated his argument with *inter alia* the responsibility and competence that goes along with owning a property right in some resource. If the owner is fully accountable, he will seek the best possible use (ibid., p. 271). Eucken accompanies the argument in favor of secure property rights immediately with the importance of competition. Only in a competitive order, can we expect the power associated with property reliably to be held in check (ibid., p. 275).
- 4. *Freedom to contract*: This point seems a straightforward requirement for competition in a market order, but again, Eucken adds some words of caution. Freedom to contract needs to be embedded in a competitive order. Naming freedom to contract as a necessary requirement for competition and at the same time the stability of a competitive order as necessary for freedom to contract to yield positive outcomes may seem paradox at first. But it is important to see that economic policy has the explicit task of resolving the paradox by seeing that the freedom to contract is not used to build cartels or reduce competition in other forms (ibid., pp. 277-279).
- 5. Accountability: One component of the principle of accountability is straightforward; individuals are accountable for their own actions. If those actions lead to profits or other benefits or to losses should not matter and, in particular, channels that allow the externalization of losses should be closed. The second component may be less obvious. Suppose that a government regulates firms heavily. If such regulations affects their competitiveness adversely, the losses accrue to shareholders, not to the political decision-makers (ibid., p. 282). The principle of accountability therefore also leads to some skepticism regarding the legitimacy of regulatory interventions in general.
- 6. Consistency of economic policy: Eucken considers it important that individuals and firms are capable of making informed decisions over longer time horizons. Such

³ See, e.g., modern Schumpeterian growth theory, as in Aghion et al. (2014).

predictability, he argues, in particularly important for stimulating investment. Thus, economic policy should be foreseeable and stable.

The constitutive principles are amended by a set of so-called regulatory principles. Eucken (1952, pp. 291-303) was well aware of the reality of market imperfections that cannot be cured by setting the rules for a competitive order alone. Natural monopolies and externalities are such cases, but he also argues that some degree of income redistribution will be necessary. Thus, the Ordoliberals have from early on reckoned that in some cases, efficiency demands direct interventions into market outcomes. But those interventions themselves are rule-based; clearly defined conditions should trigger them.⁴

All of the constitutive principles can and should be implemented by some form of rule-based policy. A quantitative monetary policy rule controlling money supply is possible, as are constitutional rules that prohibit governments from engaging in protectionist policies. Constitutional rules of non-discrimination guaranteeing that the price mechanism is not distorted by discretionary subsidies to privileged firms likewise can be conceived. Private property rights can be secured by rules that limit the sizes of permissible tax burdens. It is important to see that such rules constrain discretion and that they do not determine economic outcomes. They define a playing field and rules of the game for both private producers and consumers acting in ordinary markets and for the political process itself.

An important issue that also needs to be addressed is enforcement. The paradox that rules limiting government need to be enforced by government itself cannot be escaped. The Freiberg School's founding generation carried with them a normative idea of what they called a "strong state", i.e., a state that is not receptive to offers from special interest groups, but restricts itself to playing the role of a neutral referee. They have, however, not developed a coherent proposal as to how politicians can be bound to that normative ideal. Even though the enforcement problem remains unsolved, the idea of a strong state has drawn some criticism claiming that Ordoliberalism inherently is authoritarian.⁵

6. Rule-based policy-making and the democratic process

⁴ A referee has pointed out that the list of constitutive principles bears some resemblance to the items in the Economic Freedom of the World Index. I have, however, not found any indication that the authors of the EFW Index have been inspired directly by Eucken.

⁵ For an in-depth discussion of the issue, see Köhler and Nientiedt (2021).

It is intuitively plausible that a stable, competitive order facilitates the type of productive competition that is envisaged by Eucken as an ideal, thereby steering the economy to serve the broad interests of consumers. Something that may be less plausible is the role remaining for democratic decision-making, when the discretionary decision-making power of governments effectively is constrained. Not surprisingly, some critics of ordoliberal thinking claim that the rule-based approach reflects a thinking that is skeptical of democracy and that wants to restrain the polity to acting as a servant of an efficient market process.

The standard counterargument is that such critics of the ordoliberal position ignore that the analytical starting point for Eucken and others are the costs of not complying with their proposed principles. The historical root of the ordoliberal argument was, at the time, fresh experience of a breakdown of the economic order during the Great Depression and of Germany's political order thereafter. In their interpretation, those orders broke down as a result primarily of laissez-faire allowing market power to accumulate and of undue and inefficient discretionary policymaking that both failed to stabilize the economy and eroded the competitive order in consequence of governmental collusion with powerful special interests.

To assume that laissez-faire leads to an undue accumulation of power by special interests is not, as it may at first seem, a logical fallacy. It is a hypothesis about the emergence of power from an unchecked laissez-faire framework: Competition needs to be secured actively by the government, e.g., through active antitrust policy enforcement banning cartels and collusion. If such policies are not pursued, then, so the ordoliberal hypothesis goes, the laissez-faire order will erode itself, because market participants are allowed to become too powerful by evading competition.

The ordoliberal interpretation is pre- and post-Second World War period was not a unique historical experience, but that erosion of the competitive order associated with excessive discretionary policy is a permanent danger. The argument is therefore that only a state detached from serving special interest groups in a discretionary fashion is a strong state that can support the stability of desired orders. That is an empirical argument and it should be debated as such – we will return to this point in Section 7.

As soon as one leaves the abstract level of discussion on rule-based policymaking and discusses concrete measures, more margins of democratic decision-making appear. The first-generation members of the Freiburg School did not develop an explicit political economy of

finding and recommending a specific constitutional order for a given time and place. That gap in the theory may have contributed to a common misconception of Ordoliberalism as a doctrinal ideology that demands enforcing a fixed set of rules, specified in detail in the writings of, by now, long-deceased Freiburg School members. Clearly, that interpretation is a misrepresentation. Rules need to be negotiated and every concrete rule poses concrete tradeoffs. A political decision made along such lines reflects a society's heterogeneous values und preferences.

Economic policy proposals, from an ordoliberal or any other school, will in practice serve only as inputs into real-world democratic processes. For example, a constitutional rule that limits the scope of taxation in order to protect private property rights will in practice be implemented only after broad democratic deliberations. Those discussions could concern the allowable total tax burden or the permissible structure of a tax schedule – are, for instance, regressive income taxes acceptable? Many other features of such a tax rule need to be traded off against others. Or consider the problem of patent law raised by Eucken, who emphasizes the danger that patents will reduce competition and serve primarily as legal underpinnings of monopoly power. From a different perspective, one obviously could argue that patent law creates incentives to invest more resources in research and development and thus increases the velocity of Schumpeterian creative destruction. Again, a conflict arises between two possible objectives that can be resolved only through democratic deliberation and decision-making. One objectively correct rule for economic policymaking towards innovation policy is not available and political value judgments must be considered.

Furthermore, and little recognized in economics so far, theoretical and policy uncertainty are widespread even on issues like international trade or minimum wage laws, which have been studied for many years. Generally, no perfect *ex ante* knowledge is available about the actual functioning of an economic policy rule. And different individuals arrive at different conclusions about the effects that a rule eventually will have. Ideally, a process of thorough democratic deliberation amongst individuals who treat one another as "natural equals" (Levy and Peart 2020) facilitates an informed judgment about alternative rules. Individuals share their dispersed knowledge and a more complete, informed picture of proposed rules emerges (Vanberg and Buchanan 1989). Of course, in reality, conflicts of theory and conflicts of interest overlap; individuals therefore have different motives than to let themselves be

_

⁶ See, e.g., the proposal for constitutional tax reform by Brennan and Buchanan (1980).

persuaded by the better argument (Schnellenbach and Schubert 2015). But even if the process is not ideal, a period of democratic deliberation can improve the informational basis of a decision on rules.

The criticism that rule-based policy-making renders democratic processes irrelevant thus is grossly misleading. Rule proposals for economic policy are merely inputs into the process of democratic deliberation, intended to solve well-specified problems. A rule-based approach to economic policy is not a substitute for democratic procedure, but relies heavily on it.⁷

7. Is rule-based *Ordnungspolitik* still called for?

We have seen so far that the Freiburg School advanced plausible arguments in favor of rule-based policymaking. We likewise have seen that recent arguments that rule-based policymaking is in conflict with the principles of an open democracy are not very convincing. Historically, the catastrophic economic collapse of the Great Depression and the even more catastrophic political collapse of the Weimar democracy in Germany led the members of the Freiburg School to advocate *Ordnungspolitik*. But are other, less spectacular examples still to be found wherein rules could be useful in limiting the scope of discretionary policymaking?

Four examples can illustrate the point that, from a modern perspective, the basic claims made by the first generation of Ordoliberals remain rather robust. The examples do not include the established fields of rule-based economic policymaking, such as the adoption of fiscal rules or the delegation of monetary policy to independent central banks. Rather, they are intended to illustrate that the slow erosion of the competitive market order, that members of the Freiburg School wanted to prevent by introducing suitable rules, is still a threat to be reckoned with.

Consider, for example, the argument that individual autonomy is worthy of protection by appropriate rules. A long-standing debate in behavioral economics questions whether individuals are capable of correctly estimating *ex ante* the utilities associated with the choices they are going to make.⁸ Individuals often fail to predict expected utility accurately and therefore make choices they later regret. One could argue that such errors simply are part of

⁷ Note, however, that Eucken has been rather wary of the danger that special interest groups dominate the democratic process; see Nientiedt and Köhler (2016).

⁸ For instance, Frey and Stutzer (2007) and the discussion in Schnellenbach (2019).

the human condition, or one could propose "expert" help in making better choices. The latter approach is associated with a relatively novel strand of research on behavioral public economics, which looks for policy interventions that supposedly improve welfare by driving individuals to make choices that maximize actual, rather than falsely predicted utility. A popular aspect of that school of thought is the literature on so-called nudging, which is supposed to improve decision-making by using deliberately designed decision frames that manipulate individuals into choosing with a higher probability the alternative that a supposedly benevolent paternalist planner considers superior (e.g., Sunstein and Thaler 2003).

But fundamental empirical problems are associated with such an approach. First of all, it does not work reliably owing to an epistemic problem that is well-known: Individuals seldom know what exactly satisfies the preferences of other individuals and therefore the probability is high that interventions produce results that are not helpful (Rebonato 2012). Systematic biases also emerge. For example, in experiments, individuals who are supposed to guide the choices of others are much stricter (e.g., when it comes to a healthy diet) than they are when they consider their own choices. The result can be over-regulation of individual behavior and a loss of autonomy.

Another empirical result from experimental research shows that individuals who are subjected to paternalist interventions exhibit a positive willingness to pay to be able to make their own choices (Lusk et al. 2014; Kragh Pedersen et al. 2012). From a psychological perspective, the empirically motivated self-determination theory states that individuals generally seek autonomy in determining their own goals and become more satisfied when they can pursue them themselves. Self-determination is seen as a basic human need (Deci and Ryan 2000, 2012).

We therefore observe, on one side, well-intentioned attempts to improve individual welfare by systematically overriding consumer sovereignty. Such attempts can come in the form of proliferating consumer protection laws or in the form of nudges and other manipulative exploitations of behavioral mechanisms. On the other side, we also observe an empirically well-documented preference for individual autonomy. A good reason to value one's own autonomy highly is that individuals do not know today how their own preferences will look tomorrow. If I do not know today how my preferred consumption patterns will be tomorrow, I have little reason to favor restraining my consumer sovereignty (Sugden 2010).

A contractarian argument can then be made that individuals seek opportunity and demand an institutional order that expands and safeguards their choice sets from government intervention (Sugden 2010). That argument is indeed very close to the ordoliberal argument that a competitive economic order with an undistorted price mechanism and free of arbitrary interventions into voluntary transactions should be the general objective of rule-based policymaking. And the empirical evidence hinting at the importance of autonomy in choice also aligns with the ordoliberal stance of preserving a humane economic order, as discussed above. The preservation of consumer sovereignty by adopting appropriate rules can be supported by arguments from the modern literature on behaviorally motivated policy interventions.

As a second example, consider the costs of discretionary interventions into the competitive order. *Prima facie*, it appears that modern market economies are extremely resilient. It is safe to say that most governments do not pursue the implementation of Eucken's constitutive principles as their main policy objective. Instead, discretionary interventions into market processes are ubiquitous – but we still observe long-run growth, a reasonable degree of economic freedom, and relatively stable democracies in most advanced nations. The danger of a complete collapse and self-destruction of a competitive market order seems to be not as acute as the first generation of Ordoliberals have argued under the fresh impression of the Great Depression and rise of National Socialism, Fascism and Communism.

But although modern market economies and democracies seem to be rather resilient and can sustain high levels of productivity and social welfare even with major discretionary interventions and massive public spending; the welfare losses imposed, for example, by rent-seeking effort still are non-negligible. Even though illegal, potentially scandalous types of rent seeking typically are not very pervasive in open democracies (Hillman and Ursprung (2016), many costly rent-seeking activities are evident. Attempts arbitrarily to deter newcomers from market entry regulatorily occur on a regular basis (consider the troubles Uber has in entering many heavily regulated European taxi markets), and some countries have returned to the ancient practice of defending their home market with tariffs.

The overall effect of rent seeking on welfare in advanced economies is notoriously difficult to estimate, in particular owing to unavailable data and the interval bounding the existing estimates of rent-seeking costs as a fraction of GDP ranged from almost zero to a quarter of GDP (Del Rosal 2011). But one has to take into account that some of those costs, such as the costs of foregone innovations caused by entry barriers, escape quantification more or less

completely. In any case, rent seeking never is a productive activity and a rule-based policy that reduces rent seeking and secures the openness of markets can be expected to have positive effects, even if it is not necessary to prevent complete systemic failure of a market economy, or even an extreme accumulation of economic power that eventually threatens the functioning of democracy itself.

As a third example, consider the effects of a very generous welfare state with high replacement rates. It has been argued that in such a setting, a previously existing work ethic may be crowded out permanently (Lindbeck 1995). Indeed, some survey data align with that hypothesis. Relying on data from the World Values Survey, Lindbeck and Nyberg (2006) found that a more extensive welfare state tends to be accompanied by weaker self-reported work ethics. As far as a functioning market economy also depends on social norms and attitudes that are compatible with such an order, a danger therefore arises that very generous welfare states crowd out those very norms. The example shows that the interdependency of orders, as understood by Eucken, also can apply to informal rules and is not limited to the interactions of formal institutions.

As a final example, consider the problem of political selection. In the absence of rule-based politics, selection becomes extremely important. Once a politician is in office, the means to control her political decisions are rather imperfect. Parliamentary majorities often avoid the risks of snap elections and thus support governments, even if their policy choices run against the preferences of a significant number of members of the majority fractions. Referendums, if available at all, often are costly to organize. The more imperfect are the means of political control, the more important is the selection of politicians whose preferences align with those of a majority of voters. But what if a significant number of voters rewards trivial characteristics such as a candidate's good looks (Berggren et al 2017)? And that is only one example of what political scientists call valence.

A behavioral perspective on political decision-making processes shows that voters' choices can be driven by all kinds of considerations that have nothing to do with the material outcomes of political alternatives (Schnellenbach and Schubert 2015). If both the selection process of political representatives and the means of *ex post* control are highly imperfect, rule-based policymaking can be seen as a useful commitment device that prevents excesses

⁹ Heinemann (2008) found the result to be robust when accounting for macroeconomic influences that were omitted from the original analysis.

possibly following the accidental selection of incompetent or eccentric representatives. Modern approaches to political economics tend to confirm the imperfections of democratic procedures and highlight the importance of rule-based checks on discretionary government power. Limiting power by adopting appropriate rules is vital not only on markets in order to sustain a competitive economic order, but also in politics, where competition always is much more indirect an incomplete.

8. Conclusions

Some of the analytical starting points of the Freiburg School of economics seem peculiar from a modern perspective. In particular, market economies and modern democracies could be more robust and less vulnerable than assumed by the School's first-generation members. Under their experiences with the Great Depression and the rise of National Socialism, they argued that only a close-knit system of rules could secure the desired stability of the economic and political orders in the long run. That conclusion appears not to be the case.

Nevertheless, we argue that not having rules is costly and, hence, that the main thrust of the Freiburg approach remains valid. Good empirical arguments can be identified for pursuing a rule-based *Ordnungspolitik* in order to avoid the costs of discretionary public policymaking. Furthermore, we argue that relying on stable rules does not imply incapacitating democratic decision-making. Rules rely on democratic support and rule-based *Ordnungspolitik* also leaves substantial material scope for discretionary democratic decision-making.

While we have seen differences between the Freiburg School and the Virginia School in terms of motivating emphasis on rules, and also in terms of laying different normative foundations for rules, we also have seen overlaps. It is therefore an interesting question as to what extent Buchanan himself may have been inspired by the earlier works of Eucken and other members of the Freiburg School. It is well-established that that a main influence on Buchanan's thinking has been the "Old" Chicago School (Levy and Peart 2020). As individuals, in particular Frank Knight and Henry Simons (Köhler and Kolev 2013), who held very similar positions on the role of the state as the Ordoliberals did, played an important role in influencing Buchanan. But more importantly, it is also argued that Friedrich von Hayek was an important bridge for ideas and topics between Freiburg und Chicago, by maintaining

personal relations both to the Freiburg School, and to "Old" Chicago figures like Simons (Köhler and Kolev 2013; Kolev 2021).

Thus, there is reason to believe that the similarities in thinking that exist between the Freiburg School and the Virginia School are not entirely accidental. Certainly, Buchanan himself has emphasized repeatedly his own affinity to the tradition of thinking about orders, which both the Freiburg School and the Old Chicago School represent (Kolev 2018).

References

- Acemoglu, D., Johnson, S. and Robinson, J. A. (2001). The colonial origins of comparative development: An empirical investigation. *American Economic Review*, 91(5), 1369-1401.
- Acemoglu, D., Cantoni, D., Johnson, S. and Robinson, J.A. (2011). The consequences of radical reform: The French Revolution. *American Economic Review* 101(7): 3286-3307.
- Aghion, P., Agcigit, U. and Howitt, P. (2014). What do we learn from Schumpeterian growth theory? In: P. Aghion and S.N. Durlauf (eds.), *Handbook of economic growth*, vol. 2, pp. 515-563. Amsterdam: Elsevier.
- Berggren, N., Jordahl, H. and Poutvaara, P. (2017). The right look: Conservative politicians look better and voters reward it. *Journal of Public Economics* 146: 79-86.
- Böhm, F. (1937). Die Ordnung der Wirtschaft als geschichtliche Aufgabe und rechtsschöpferische Leistung. Stuttgart: Kohlhammer.
- Böhm, F. (1966). Privatrechtsgesellschaft und Marktwirtschaft. Ordo Jahrbuch für die Ordnung von Wirtschaft und Gesellschaft 17: 75-151.
- Böhm, F. (1971). Freiheit in Ordnung in der Marktwirtschaft. Reprinted in N. Goldschmidt and M. Wohlgemuth (2008), pp. 299-312.
- Böhm, F., Eucken, W. and Großmann-Doerth, H. (1936). Unsere Aufgabe. Reprinted in N. Goldschmidt and M. Wohlgemuth (2008), pp. 27-41.
- Brennan, B. and Buchanan, J.M. (1980). *The power to tax. The analytical foundations of a fiscal constitution*. Cambridge: Cambridge University Press.
- Deci, E.L. and Ryan, R.M. (2000). The 'what' and 'why' of goal pursuits: Human needs and the self-determination of behavior, *Psychological Inquiry*. 11: 227-268.
- Deci, E.L. and Ryan, R.M. (2012). Motivation, personality and development within embedded social contexts: An overview of self-determination theory, in: R.M. Ryan (ed.). *Oxford handbook of human motivation*. Oxford: Oxford University Press: 85-107.
- Del Rosal, I. (2011). The empirical measurement of rent-seeking Costs. *Journal of Economic Surveys* 25(2): 298-325.
- Denzau, A.T. and North, D.C. (1994). Shared mental models: Ideologies and institutions. Kyklos 47(1): 3-31.
- Eucken, W. (1939). Grundlagen der Nationalökonomie. Jena: Fischer.
- Eucken, W. (1947). Über die zweifache wirtschaftspolitische Aufgabe der Nationalökonomie. Reprinted in N. Goldschmidt and M. Wohlgemuth (2008), pp. 133-154.
- Eucken, W. (1952). Grundsätze der Wirtschaftspolitik. 7th ed., Tübingen: Mohr Siebeck.

- Feld, L.P. and Köhler, E.A. (2011). Ist die Ordnungsökonomik zukunftsfähig? *Zeitschrift für Wirtschafts- und Unternehmensethik* 12: 173-195.
- Feld, L.P., Köhler, E.A. and Nientiedt, D. (2021). Die Europäische Währungsunion aus traditioneller und moderner ordnungsökonomischer Perspektive. *Ordo Jahrbuch für die Ordnung von Wirtschaft und Gesellschaft* 69, forthcoming.
- Frey, B.S. and Stutzer, A. (2007). What happiness research can tell us about self-control problems and utility misprediction. In: B.S. Frey and A. Stutzer (eds.), *Economics and psychology. A promising new cross-disciplinary field*, pp. 169-196. Cambridge (Mass.): MIT Press.
- Goldschmidt, N. and Wohlgemuth, M. (eds.) (2008). *Grundtexte zur Freiburger Tradition der Ordnungsökonomik*. Untersuchungen zur Ordnungstheorie und Ordnungspolitik 50. Tübingen: Mohr Siebeck.
- Heinemann, F. (2008). Is the welfare state self-destructive? A study of government benefit morale. *Kyklos* 61: 237-257.
- Hillman, A.L. and Ursprung, H. (2016). Where are the rent-seekers? *Constitutional Political Economy* 27(2): 124-141.
- Köhler, E.A. and Kolev, S. (2013). The conjoint quest for a liberal positive program: "Old Chicago", Freiburg, and Hayek. In: S.J. Peart and D.M. Levy (eds.), *F.A. Hayek and the modern economy*, pp. 211-228, New York, Palgrave Macmillan.
- Köhler, E.A. and Nientiedt, D. (2021). Was Walter Eucken a proponent of authoritarian liberalism? *Public Choice*, this volume.
- Köhler, E.A. and Vanberg, V.J. (2015). The constitutionalization of money: A constitutional economics perspective. In: L.H. White, V.J. Vanberg and E.A. Köhler (eds.), *Renewing the search for a monetary constitution*, pp. 59-103, Washington, DC: CATO Institute.
- Kolev, S. (2018). James Buchanan and the "new economics of order" research program. In: Richard E. Wagner (ed.), *James M. Buchanan. A theorist of political economy and social philosophy*, pp- 85-108, New York: Palgrave Macmillan.
- Kolev, S. (2021). When liberty presupposes order: F.A. Hayek's learning ordoliberalism. Freiburg Discussion Papers on Constitutional Economics 21/02, Freiburg: Walter Eucken Institut.
- Kolev, S. (2020). Ordoliberalism's embeddedness in the neoliberalisms of the 1930s and 1940s. In: M. Dold and T. Krieger (eds.), *Ordoliberalism and European economic policy: Between Realpolitik and economic utopia*, pp. 23-38, London: Routledge.
- Kragh Pedersen, S., Koch, A.K. and Nafziger, J.- (2014). Who wants paternalism?, *Economics Bulletin*. 66: S147-S166.
- Levy, D.M. and Peart, S. (2020). *Towards and economics of natural equals. A documentary history of the Virginia School*. Cambridge: Cambridge University Press.
- Lindbeck, A. (1995). Hazardous welfare-state dynamics. American Economic Review (P&P) 85: 9-15.
- Lindbeck, A. and Nyberg, S. (2006). Raising children to work hard: altruism, work norms and social insurance. *Quarterly Journal of Economics* 121: 1473-1503.
- Lusk, J.L., Marette, S. and Norwood, F.B. (2014). The paternalist meets his match, *Applied Economic Perspectives and Policy* 36: 61-108.
- Nientiedt, D. and Köhler, E.A. (2016). Liberalism and democracy. A comparative reading of Eucken and Hayek. *Cambridge Journal of Economics* 40: 1743-1760.
- North, D.C. (1991). Institutions. *Journal of Economic Perspectives* 5(1): 97-112.
- North, D.C. (1994). Economic performance through time. American Economic Review 84(3): 359-368.
- North, D.C. and Weingast, B.R. (1989). Constitutions and commitment: The evolution of institutions governing Public Choice in seventeenth-century England. *Journal of Economic History* 49(4): 803-832.
- Rebonato, R. (2012). Taking liberties. A critical examination of libertarian paternalism. Basingstoke: Palgrave Macmillan.
- Röpke, W. (1944). Civitas Humana. Grundfragen der Gesellschafts- und Wirtschaftsreform. Erlenbach ZH: Rentsch.

- Schnellenbach, J. (2019). Evolving hierarchical preferences and behavioral economic policies. *Public Choice* 178: 31-52.
- Schnellenbach, J. and Schubert, C. (2015). Behavioral political economy. *European Journal of Political Economy* 40: 395-417.
- Sugden, R. (2010). Opportunity as mutual advantage. Economics & Philosophy 26: 47-68.
- Sunstein, C.R. and Thaler, R.H. (2003). Libertarian paternalism is not an oxymoron. *University of Chicago Law Review* 70: 1159-1202.
- Vanberg, V.J. (1988). 'Ordnungstheorie' as constitutional economics. The German conception of a 'social market economy'. *Ordo Jahrbuch für die Ordnung von Wirtschaft und Gesellschaft* 39: 17-31.
- Vanberg, V.J. (2004). *The Freiburg School: Walter Eucken and Ordoliberalism*. Freiburg Discussion Papers on Constitutional Economics 04/11. Freiburg: Walter Eucken Institut.
- Vanberg, V.J. (2015). Ordoliberalism, Ordnungspolitik and the reason of rules. *European Review of International Studies* 25: 27-36.
- Vanberg, V.J. and Buchanan, J.M. (1989). Interests and theories in constitutional choice. *Journal of Theoretical Politics* 1: 49-62.
- Williamson, O. (1985). The economic institutions of capitalism. New York: The Free Press.

Freiburger **Diskussionspapiere** zur Ordnungsökonomik

Freiburg **Discussion Papers** on Constitutional Economics

2021							
21/7	Jan Schnellenbach: The Concept of Ordnungspolitik: Rule-Based Economic Policy-Making from the Perspective of the Freiburg School						
21/6	Jan Schnellenbach: Herausforderungen für die Finanzpolitik in Deutschland nach der Covid-Krise: Schuldenbremse und Vermögensbesteuerung						
21/5	Lars P. Feld / Ekkehard A. Köhler / Daniel Nientiedt: Ordoliberalism and the Social Market Economy						
21/4	Bury Yannick / Feld, Lars P. / Burret, Heiko T.: Skimming the Achieved? – Quantifying the Fiscal Incentives of the German Fiscal Equalization Scheme and its Reforms since 1970						
21/3	Le Maux, Benoît / Masclet, David / Necker, Sarah: Monetary Incentives and the Contagion of Unethical Behavior						
21/2	Kolev, Stefan: When Liberty Presupposes Order: F. A. Hayek's Learning Ordoliberalism						
21/1	Burret, Heiko T. / Feld, Lars P. / Schaltegger, Christoph A.: Fiscal Federalism and Economic Performance – New Evidence from Switzerland						
2020							
20/10	Christofzik, Désirée / Feld, Lars P. / Yeter, Mustafa: Heterogeneous Price and Quantity Effects of the Real Estate Transfer Tax in Germany						
20/9	Hirsch, Patrick / Köhler, Ekkehard A. / Feld, Lars P. / Thomas, Tobias: "Whatever It Takes!" How Tonality of TV-News Affects Government Bond Yield Spreads During Crises						
20/8	Feld, Lars P.: Die wirtschaftlichen Rahmenbedingungen nach dem Corona-Schock						
20/7	Feld, Lars P.: Wohlstand für alle – Was das Versprechen heute bedeutet. Ein Vorwort						
20/6	Feld, Lars P.: Wirtschaftskrisen der Zukunft						
20/5	Feld, Lars P. / Wieland, Volker: The German Federal Constitutional Court Ruling and the European Central Bank's Strategy						
20/4	Bury, Yannick / Feld, Lars P.: Fiscal Federalism in Germany						
20/3	Vanberg, Viktor J.: Walter Euckens Weg zum Ordoliberalismus						
20/2	Feld, Lars P.: Verfahren zum Anleihekaufprogramm der EZB(2 BvR 859/15, 2 BvR 1651/15, 2 BvR 2006/15, 2 BvR 980/16). Stellungnahmezum Fragenkatalog für sachverständige Dritte						
20/1	Feld, Lars P. / Reuter, Wolf Heinrich / Yeter, Mustafa: Öffentliche Investitionen: Die Schuldenbremse ist nicht das Problem						
2019							
19/6	Kolev, Stefan: Antipathy for Heidelberg, Sympathy for Freiburg? Vincent Ostrom on Max Weber, Walter Eucken, and the Compound History of Order						
19/5	Feld, Lars P. / Frey, Christian / Schaltegger, Christoph A. / Schmid, Lukas A.: Fiscal Federalism and Income Inequality: An Empirical Analysis for Switzerland						
19/4	Goldschmidt, Nils / Wolf, Stephan: Klimaschutz auf Kosten der Armen? Vorschläge für eine markt- und sozialverträgliche Umsetzung von CO2-Steuern und des Emissionshandels						
19/3	Horn, Karen I.: The Difficult Relationship Between Historical Ordoliberalism and Adam Smith						
19/2	Christofzik, Désiree / Feld, Lars P. / Yeter, Mustafa: Öffentliche Investitionen: Wie viel ist zu wenig?						
19/1	Feld, Lars P. / Hirsch, Patrick: Zur Grundsteuerreform						

- **18/13 Doerr, Anabelle / Necker, Sarah:** Toward an Understanding of Collaborative Tax Evasion: A Natural Field Experiment With Businesses
- **Bury, Yannick / Feld, Lars P.:** Die Heterogenität der kommunalen Haushalts- und Aufsichtsregeln als Herausforderung im vertikalisierten Fiskalföderalismus
- **18/11 Feld, Lars P. / Köhler, Ekkehard A. / Nientiedt, Daniel:** The German Anti-Keynes? On Walter Eucken's Macroeconomics
- **Dathe, Uwe / Hedtke, Ulrich:** Habent sua fata professores. Joseph A. Schumpeter an Walter Eucken
- 18/09 Feld, Lars P.: The Quest for Fiscal Rules
- **18/08 Pfeil, Christian F. / Feld, Lars P.:** Does the Swiss Debt Brake Induce Sound Federal Finances? A Synthetic Control Analysis
- **18/07 Feld, Lars P.:** Zur Politischen Ökonomik der wirtschaftspolitischen Beratung: Der Sachverständigenrat als ordnungspolitisches Gewissen?
- **18/06** Koessler, Ann-Kathrin / Torgler, Benno / Feld, Lars P. / Frey, Bruno S: Commitment to Pay Taxes: Results from Field and Laboratory Experiments
- **18/05 Feld, Lars P. / Köhler, Ekkehard A. / Wolfinger, Julia:** Modeling Fiscal Sustainability in Dynamic Macro-Panels with Heterogeneous Effects: Evidence From German Federal States
- 18/04 Burret, Heiko T. / Bury, Yannick / Feld, Lars P.: Grenzabschöpfungsraten im deutschen Finanzausgleich
- **18/03 Vanberg, Viktor J.:** Individual Choice and Social Welfare. Theoretical Foundations of Political Economy
- 18/02 Feld, Lars P. / Schaltegger, Christoph A. / Studerus, Janine: Regional Risk Sharing and Redistribution the Role of Fscal Mechanisms in Switzerland
- 18/01 Andritzki, Jochen / Christofzik, Désiree I. / Feld, Lars P. / Scheuering, Uwe: A Mechanism to Regulate Sovereign Debt Restructuring in the Euro Area

- 17/03 Feld, Lars P. / Köhler, Ekkehard A. / Nientiedt, Daniel: The "Dark Ages" of German Macroeconomics and Other Alleged Shortfalls in German Economic Thought
- 17/02 Doerr, Annabelle: Back to work: The Long-term Effects of Vocational Training for Female Job Returners*
- 17/01 Baskaran, Thushyanthan / Feld, Lars P. / Necker, Sarah: Depressing dependence? Transfers and economic growth in the German States, 1975-2005

- **16/08 Fitzenberger, Bernd / Furdas, Marina / Sajons, Christoph:** End-of-Year Spending and the Long-Run Employment Effects of Training Programs for the Unemployed
- 16/07 Sajons, Christoph: Birthright Citizenship and Parental Labor Market Integration
- 16/06 Pfeil, Christian F.: Electoral System Change and Spending: Four Quantitative Case Studie
- 16/05 Sajons, Christoph: Information on the ballot, voter satisfaction and election turnout
- 16/04 Vanberg, Viktor J.: Social Contract vs. Invisible Hand: Agreeing to Solve Social Dilemmas
- 16/03 Feld, Lars P. / Ruf, Martin / Schreiber, Ulrich / Todtenhaupt, Maximilian / Voget, Johannes: Taxing Away M&A: The Effect of Corporate Capital Gains Taxes on Acquisition Activity
- **16/02 Baskaran, Thushyanthan / Feld, Lars P. / Schnellenbach, Jan**: Fiscal Federalism, Decentralization and Economic Growth: A Meta-Analysis
- 16/01 Burret, Heiko T. / Feld, Lars P.: Vertical Effects of Fiscal Rules The Swiss Experience

- 15/11 Burret, Heiko T. / Feld, Lars P. / Köhler, Ekkehard A.: Fiscal Sustainability of the German Laender. Time Series Evidence
- 15/10 Feld, Lars P. / Fritz, Benedikt: The Political Economy of Municipal Amalgamation. Evidence of Common Pool Effects and Local Public Debt
- **Burret, Heiko T. / Feld, Lars P. / Köhler, Ekkehard A.:** (Un-)Sustinability of Public Finances in German Laender. A Panel Time Series Approach
- **15/8 Feld, Lars P. / Köhler, Ekkehard A.:** Is Switzerland an Interest Rate Island After All? Time Series and Non-Linear Switching Regime Evidence
- **Doerr, Annabelle / Fitzenberger, Bernd:** Konzeptionelle Lehren aus der ersten Evaluationsrunde der Branchenmindestlöhne in Deutschland
- 15/6 Vanberg, Viktor J.: Constitutional Political Economy
- 15/5 Vanberg, Viktor J.: Competitive Federalism, Government's Dual Role, and the Power to Tax
- 15/4 Feld, Lars P. / Köhler, Ekkehard A. / Nientiedt, Daniel: Ordoliberalism, Pragmatism and the Eurozone Crisis: How the German Tradition Shaped Economic Policy in Europe
- **Vanberg, Viktor:** "Freiheit statt Kapitalismus?" Ein Kommentar zu Sahra Wagenknechts Buch aus Freiburger Sicht
- 15/2 Schnellenbach, Jan: A Constitutional Economics Perspective on Soft Paternalism
- **Schnellenbach, Jan:** Die Politische Ökonomie des Entscheidungsdesigns: Kann Paternalismus liberal sein?
- 2014
- **14/8 Schnellenbach, Jan:** Neuer Paternalismus und individuelle Rationalität: eine ordnungsökonomische Perspektive
- 14/7 Schnellenbach, Jan: Does Classical Liberalism Imply an Evolutionary Approach to Policy-Making?
- **14/6 Feld, Lars P.:** James Buchanan's Theory of Federalism: From Fiscal Equity to the Ideal Political Order
- 14/5 Reckendrees, Alfred: Weimar Germany: the First Open Access Order that Failed
- 14/4 Vanberg, Viktor J.: Liberalismus und Demokratie. Zu einer vernachlässigten Seite der liberalen Denktradition
- 14/3 Schnellenbach, Jan / Schubert, Christian: Behavorial Public Choice: A Survey
- 14/2 Goldschmidt, Nils / Hesse, Jan-Otmar / Kolev, Stefan: Walter Eucken's Role in the Early History of the Mont Pèlerin Society
- 14/1 Vanberg, Viktor J.: Ordnungspolitik, the Freiburg School and the Reason of Rules
- 2013
- **Wegner, Gerhard:** Capitalist Transformation Without Political Participation German Capitalism in the First Half of the 19th Century
- 13/13 Necker, Sarah / Voskort, Andrea: The Evolution of Germans` Values since Reunification
- 13/12 Biedenkopf, Kurt: Zur ordnungspolitischen Bedeutung der Zivilgesellschaft
- 13/11 Feld, Lars P. / Ruf, Martin / Scheuering, Uwe / Schreiber, Ulrich / Voget, Johannes: Effects of Territorial and Worldwide Corporation Tax Systems on Outbound M&As
- 13/10 Feld, Lars P. / Kallweit, Manuel / Kohlmeier, Anabell: Maßnahmen zur Vermeidung von Altersarmut: Makroökonomische Folgen und Verteilungseffekte
- **Feld, Lars P.:** Zur Bedeutung des Manifests der Marktwirtschaft oder: Das Lambsdorff-Papier im 31. Jahr.
- 13/8 Feld, Lars P. / Köhler, Ekkehard A.: Is Switzerland After All an Interest Rate Island?
- 13/7 Feld, Lars P. / Necker, Sarah / Frey, Bruno S.: Happiness of Economists

- **13/6 Feld, Lars P. / Schnellenbach, Jan:** Political Institutions and Income (Re-)Distribution: Evidence from Developed Economies
- 13/5 Feld, Lars P. / Osterloh, Steffen: Is a Fiscal Capacity Really Necessary to Complete EMU?
- 13/4 Vanberg, Viktor J.: James M. Buchanan's Contractarianism and Modern Liberalism
- 13/3 Vanberg, Viktor J.: Föderaler Wettbewerb, Bürgersouveränität und die zwei Rollen des Staates
- 13/2 Bjørnskov, Christian / Dreher, Axel / Fischer, Justina A.V. / Schnellenbach, Jan / Gehring, Kai: Inequality and happiness: When perceived social mobility and economic reality do not match
- 13/1 Mayer, Thomas: Die Ökonomen im Elfenbeinturm: ratlos Eine österreichische Antwort auf die Krise der modernen Makroökonomik und Finanztheorie

- 12/5 Schnellenbach, Jan: The Economics of Taxing Net Wealth: A Survey of the Issues
- 12/4 Goldschmidt, Nils / Hesse, Jan-Otmar: Eucken, Hayek, and the Road to Serfdom
- **Goldschmidt, Nils:** Gibt es eine ordoliberale Entwicklungsidee? Walter Euckens Analyse des gesellschaftlichen und wirtschaftlichen Wandels
- 12/2 Feld, Lars P.: Europa in der Welt von heute: Wilhelm Röpke und die Zukunft der Europäischen Währungsunion
- 12/1 Vanberg, Viktor J.: Hayek in Freiburg

2011

- 11/4 Leuermann, Andrea / Necker, Sarah: Intergenerational Transmission of Risk Attitudes A Revealed Preference Approach
- 11/3 Wohlgemuth, Michael: The Boundaries of the State
- 11/2 Feld, Lars P. / Köhler Ekkehard A.: Zur Zukunft der Ordnungsökonomik
- 11/1 Vanberg, Viktor J.: Moral und Wirtschaftsordnung: Zu den ethischen Grundlagen einer freien Gesellschaft

2010

- 10/5 Bernholz, Peter: Politics, Financial Crisis, Central Bank Constitution and Monetary Policy
- 10/4 Tietmeyer, Hans: Soziale Marktwirtschaft in Deutschland Entwicklungen und Erfahrungen
- 10/3 Vanberg, Viktor J.: Freiheit und Verantwortung: Neurowissenschaftliche Erkenntnisse und ordnungsökonomische Folgerungen
- 10/2 Vanberg, Viktor J.: Competition among Governments: The State's Two Roles in a Globalized World
- 10/1 Berghahn, Volker: Ludwig Erhard, die Freiburger Schule und das 'Amerikanische Jahrhundert'

- **09/10 Dathe, Uwe:** Walter Euckens Weg zum Liberalismus (1918-1934)
- 09/9 Wohlgemuth, Michael: Diagnosen der Moderne: Friedrich A. von Hayek
- 09/8 Bernhardt, Wolfgang: Wirtschaftsethik auf Abwegen
- 09/7 Mäding, Heinrich: Raumplanung in der Sozialen Marktwirtschaft: Ein Vortrag
- 09/6 Koenig, Andreas: Verfassungsgerichte in der Demokratie bei Hayek und Posner
- 09/5 Berthold, Norbert / Brunner, Alexander: Gibt es ein europäisches Sozialmodell?
- 09/4 Vanberg, Viktor J.: Liberal Constitutionalism, Constitutional Liberalism and Democracy
- **Vanberg, Viktor J.:** Consumer Welfare, Total Welfare and Economic Freedom On the Normative Foundations of Competition Policy
- 09/2 Goldschmidt, Nils: Liberalismus als Kulturideal. Wilhelm Röpke und die kulturelle Ökonomik.
- **09/1 Bernhardt, Wolfgang:** Familienunternehmen in Zeiten der Krise Nachhilfestunden von oder für Publikumsgesellschaften?

- **08/10** Borella, Sara: EU-Migrationspolitik. Bremse statt Motor der Liberalisierung.
- **Wohlgemuth, Michael:** A European Social Model of State-Market Relations: The ethics of competition from a "neo-liberal" perspective.
- **Vanberg, Viktor J.:** Markt und Staat in einer globalisierten Welt: Die ordnungsökonomische Perspektive.
- **Vanberg, Viktor J.:** Rationalität, Regelbefolgung und Emotionen: Zur Ökonomik moralischer Präferenzen. Veröffentlicht in: V. Vanberg: Wettbewerb und Regelordnung, Tübingen: Mohr, 2008, S. 241-268.
- **Vanberg, Viktor J.:** Die Ethik der Wettbewerbsordnung und die Versuchungen der Sozialen Marktwirtschaft
- 08/5 Wohlgemuth, Michael: Europäische Ordnungspolitik
- **08/4 Löwisch, Manfred:** Staatlicher Mindestlohn rechtlich gesehen Zu den gesetzgeberischen Anstrengungen in Sachen Mindestlohn
- 08/3 Ott, Notburga: Wie sichert man die Zukunft der Familie?
- **Vanberg, Viktor J.:** Schumpeter and Mises as 'Austrian Economists'
- **08/1** Vanberg, Viktor J.: The 'Science-as-Market' Analogy: A Constitutional Economics Perspective.

- **Wohlgemuth, Michael:** Learning through Institutional Competition. Veröffentlicht in: A. Bergh und R. Höijer (Hg.). Institutional Competition, Cheltenham: Edward Elgar, 2008, S. 67-89.
- **O7/8 Zweynert, Joachim:** Die Entstehung ordnungsökonomischer Paradigmen theoriegeschichtliche Betrachtungen.
- 07/7 Körner, Heiko: Soziale Marktwirtschaft. Versuch einer pragmatischen Begründung.
- **Vanberg, Viktor J.:** Rational Choice, Preferences over Actions and Rule-Following Behavior.
- **Vanberg, Viktor J.:** Privatrechtsgesellschaft und ökonomische Theorie. Veröffentlicht in: K. Riesenhuber (Hg.) Privatrechtsgesellschaft Entwicklung, Stand und Verfassung des Privatrechts, Tübingen: Mohr Siebeck, 2008, S. 131-162.
- **O7/4 Goldschmidt, Nils / Rauchenschwandtner, Hermann:** The Philosophy of Social Market Economy: Michel Foucault's Analysis of Ordoliberalism.
- **07/3 Fuest, Clemens:** Sind unsere sozialen Sicherungssysteme generationengerecht?
- **07/2 Pelikan, Pavel:** Public Choice with Unequally Rational Individuals.
- **Voßwinkel, Jan:** Die (Un-)Ordnung des deutschen Föderalismus. Überlegungen zu einer konstitutionenökonomischen Analyse.

- **O6/10 Schmidt, André:** Wie ökonomisch ist der "more economic approach"? Einige kritische Anmerkungen aus ordnungsökonomischer Sicht.
- **Vanberg, Viktor J.:** Individual Liberty and Political Institutions: On the Complementarity of Liberalism and Democracy. Veröffentlicht in: Journal of Institutional Economics, Vol. 4, Nr. 2, 2008, S. 139-161.
- **Goldschmidt, Nils:** Ein "sozial temperierter Kapitalismus"? Götz Briefs und die Begründung einer sozialethisch fundierten Theorie von Markt und Gesellschaft. Veröffentlicht in: Freiburger Universitätsblätter 42, Heft 173, 2006, S. 59-77.
- **Wohlgemuth, Michael / Brandi, Clara:** Strategies of Flexible Integration and Enlargement of the European Union. A Club-theoretical and Constitutional Economics Perspective. Veröffentlicht in: Varwick, J. / Lang. K.O. (Eds.): European Neighbourhood Policy, Opladen: Budrich, 2007, S. 159-180.

- **Vanberg, Viktor J.:** Corporate Social Responsibility and the "Game of Catallaxy": The Perspective of Constitutional Economics. Veröffentlicht in: Constitutional Political Economy, Vol. 18, 2007, S. 199-222.
- **Pelikan, Pavel:** Markets vs. Government when Rationality is Unequally Bounded: Some Consequences of Cognitive Inequalities for Theory and Policy.
- **Goldschmidt, Nils:** Kann oder soll es Sektoren geben, die dem Markt entzogen werden und gibt es in dieser Frage einen (unüberbrückbaren) Hiatus zwischen 'sozialethischer' und 'ökonomischer' Perspektive? Veröffentlicht in: D. Aufderheide, M. Dabrowski (Hrsg.): Markt und Wettbewerb in der Sozialwirtschaft. Wirtschaftsethische Perspektiven für den Pflegesektor, Berlin: Duncker & Humblot 2007, S. 53-81.
- **Marx, Reinhard:** Wirtschaftsliberalismus und Katholische Soziallehre.
- **Vanberg, Viktor J.:** Democracy, Citizen Sovereignty and Constitutional Economics. Veröffentlicht in: Constitutional Political Economy Volume 11, Number 1, März 2000, S. 87-112 und in: Casas Pardo, J., Schwartz, P.(Hg.): Public Choice and the Challenges of Democracy, Cheltenham: Edward Elgar, 2007, S. 101-120.
- **Wohlgemuth, Michael:** Demokratie und Marktwirtschaft als Bedingungen für sozialen Fortschritt. Veröffentlicht in: R. Clapham, G. Schwarz (Hrsg.): Die Fortschrittsidee und die Marktwirtschaft, Zürich: Verlag Neue Zürcher Zeitung 2006, S. 131-162.

- **Kersting, Wolfgang:** Der liberale Liberalismus. Notwendige Abgrenzungen. In erweiterter Fassung veröffentlicht als: Beiträge zur Ordnungstheorie und Ordnungspolitik Nr. 173, Tübingen: Mohr Siebeck 2006.
- Vanberg, Viktor J.: Der Markt als kreativer Prozess: Die Ökonomik ist keine zweite Physik. Veröffentlicht in: G. Abel (Hrsg.): Kreativität. XX. Deutscher Kongress für Philosophie. Kolloquiumsbeiträge, Hamburg: Meiner 2006, S. 1101-1128.
- **Vanberg, Viktor J.:** Marktwirtschaft und Gerechtigkeit. Zu F.A. Hayeks Kritik am Konzept der "sozialen Gerechtigkeit". Veröffentlicht in: Jahrbuch Normative und institutionelle Grundfragen der Ökonomik, Bd. 5: "Soziale Sicherung in Marktgesellschaften", hrsg. von M. Held, G. Kubon-Gilke, R. Sturn, Marburg: Metropolis 2006, S. 39-69.
- **Goldschmidt, Nils:** Ist Gier gut? Ökonomisches Selbstinteresse zwischen Maßlosigkeit und Bescheidenheit. Veröffentlicht in: U. Mummert, F.L. Sell (Hrsg.): Emotionen, Markt und Moral, Münster: Lit 2005, S. 289-313.
- **Wohlgemuth, Michael:** Politik und Emotionen: Emotionale Politikgrundlagen und Politiken indirekter Emotionssteuerung. Veröffentlicht in: U. Mummert, F.L. Sell (Hrsg.): Emotionen, Markt und Moral, Münster: Lit 2005, S. 359-392.
- 05/8 Müller, Klaus-Peter / Weber, Manfred: Versagt die soziale Marktwirtschaft? Deutsche Irrtümer.
- **Borella, Sara:** Political reform from a constitutional economics perspective: a hurdle-race. The case of migration politics in Germany.
- **05/6 Körner, Heiko:** Walter Eucken Karl Schiller: Unterschiedliche Wege zur Ordnungspolitik.
- **Vanberg, Viktor J.:** Das Paradoxon der Marktwirtschaft: Die Verfassung des Marktes und das Problem der "sozialen Sicherheit". Veröffentlicht in: H. Leipold, D. Wentzel (Hrsg.): Ordnungsökonomik als aktuelle Herausforderung, Stuttgart: Lucius & Lucius 2005, S. 51-67.
- **Weizsäcker, C. Christian von:** Hayek und Keynes: Eine Synthese. In veränderter Fassung veröffentlicht in: ORDO, Bd. 56, 2005, S. 95-111.
- **Zweynert, Joachim / Goldschmidt, Nils:** The Two Transitions in Central and Eastern Europe and the Relation between Path Dependent and Politically Implemented Institutional Change. In veränderter Fassung veröffentlicht in: Journal of Economic Issues, Vol. 40, 2006, S. 895-918.
- **Vanberg, Viktor J.:** Auch Staaten tut Wettbewerb gut: Eine Replik auf Paul Kirchhof. Veröffentlicht in: ORDO, Bd. 56, 2005, S. 47-53.
- **05/1 Eith, Ulrich / Goldschmidt, Nils:** Zwischen Zustimmungsfähigkeit und tatsächlicher Zustimmung: Kriterien für Reformpolitik aus ordnungsökonomischer und politikwissenschaftlicher Perspektive. Veröffentlicht in: D. Haubner, E. Mezger, H. Schwengel (Hrsg.): Agendasetting und

	Reformpolitik. 2005, S. 51-70.	Strategische	Kommunikatio	n zwischen v	verschiedenen	Welten, Marbur	g: Metropolis
Eine Aut	fstellung über w	veitere Disk	ussionspapiere	e ist auf der	Homepage c	les Walter Eucl	ken Instituts